\begin{table}%t1
%\centering
\par
\caption{\label{tab:HIres}Physical properties of Malin~1 and NGC~7589.}
\begin{tabular}{l c c c}
\hline \hline\noalign{\smallskip}
Property & Malin~1 & NGC 7589 & Ref.\\
\hline
Morphological Type    & SB0/a     & SAB(rs)a        & 1, 2\\
AGN activity          & LINER   &  BLAGN          & 1, 2\\
$M_{\rm{B}}$ (mag)    & $-22.1 \pm 0.4 $ & $-20.4 \pm 0.2$ & 3\\
Redshift              & $0.0826 \pm 0.0017$ & $0.0298 \pm 0.0017$ & 4\\
$D_{\rm{L}}$ (Mpc)         & $377 \pm 8$ & $130 \pm 8$ & 4\\
$D_{\rm{A}}$ (Mpc)         & $322 \pm 6$ & $123 \pm 7$ & 4\\
\hi~flux (Jy~km~s$^{-1}$)  & $2.0 \pm 0.2$ & $3.8 \pm 0.3$ & 4\\
$M_{\rm{\hi~}}$ ($10^{10}~M_{\odot}$)    & $6.7 \pm 1.0$ & $1.5 \pm 0.3$ & 4\\
SFR$_{\rm{FUV}}$ ($M_{\odot}$~yr$^{-1}$) & 1.2 & 0.7 & 5 \\
SFR$_{\rm{NUV}}$ ($M_{\odot}$~yr$^{-1}$) & 2.5 & 1.1 & 5 \\
\hline
\end{tabular}
\tablebib{(1)~\citet{Barth2007}; (2)~Nasa Extra-galactic Database (NED); (3)~\citet{Pickering1997} give $M_{\rm{V}} = -22.9 \pm 0.4$ for Malin~1 and $M_{R} = -21.9 \pm 0.2$ for NGC~7589; \citet{Impey1989} give $B - V = 0.8$ for Malin~1; \citet{Galaz2006} give $B-R = 1.5$ for NGC~7589; (4)~This work; (5)~\citet{Boissier2008}.}
\vspace*{-2mm}
\end{table}